PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

Title (Provisional)

Investigating biographical post-bariatric surgery uncertainties in the light of changes in bodily practices: a mixed-method, multicentric and longitudinal research protocol

Authors

Ferez, Sylvain; Bouchet-Mayer, Cyriac; Charissou-Pujol, Lise; Terral, Philippe; Couvry, Camille; Issanchou, Damien; Julien, Marie-Pierre; Perrin, Claire; Oppert, Jean-Michel; Ritz, Patrick; Disse, Emmanuel; Study group, Members

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.